Author's response to reviews

Title: Lack of evidence of viral reactivation in HBsAg-negative HBcAb-positive and HCV patients undergoing immunosuppressive therapy for psoriasis.

Authors:

Filomena Morisco (filomena.morisco@unina.it)
Maria Guarino (maria.guarino86@gmail.com)
Serena La Bella (serena.labella@yahoo.it)
Luisa Di Costanzo (morisco@unina.it)
Nicola Caporaso (nicola.caporaso@unina.it)
Fabio Ayala (ayala@unina.it)
Nicola Balato (nicola.balato@unina.it)

Version: 2  Date: 10 September 2014

Author's response to reviews:

made.